Author’s response to reviews

Title: Using four different clinical tools as predictors for pain after total hip arthroplasty. A prospective cohort study

Authors:

Anja Geisler (agei@regionsjaelland.dk)
Josephine Zachodnik (jzc@regionsjaelland.dk)
Jens Laigaard (jelai@regionsjaelland.dk)
Laura Kruuse (kruuse9@msn.com)
Charlotte Sørensen (cvs@regionsjaelland.dk)
Magnus Sandberg (magnus.sandberg@med.lu.se)
Eva Persson (eva_i.persson@med.lu.se)
Ole Mathiesen (omat@regionsjaelland.dk)

Version: 2 Date: 04 Feb 2020

Author’s response to reviews:

Hiroyuki Seki (Reviewer 2): P 15, Line 21-25 should be deleted, as the authors included planned number of patients as they stated.

Author’s answer:

The lines have been erased